



Brent Neumann, PhD

Group Leader/Senior Research Fellow
Neuroscience Program, Monash Biomedicine Discovery Institute and Department of
Anatomy and Developmental Biology
Monash University
Molbourne VIC 3800, Australia

Melbourne VIC 3800, Australia

Ph: +61 3 9905 0670

Email: brent.neumann@monash.edu

http://www.neumannlab.com

16th April 2019

Dear Dr Wu,

We are submitting a revised version of manuscript JoVE59978_R1, entitled "Quantitative approaches for studying cellular structures and organelle morphology in *C. elegans*".

We have addressed the second round of editorial comments and made the appropriate revisions to our manuscript. All changes have been tracked in the submitted document, and included here below is a point-by-point response to the each of the comments received.

We trust that our manuscript will now be suitable for publication in the *Journal of Visualized Experiments*.

Sincerely,

Brent Neumann

Editorial comments

The manuscript has been modified and the updated manuscript, 59978_R1.docx, is attached and located in your Editorial Manager account. Please use the updated version to make your revisions.

All changes have been made using the modified version.

1. Please take this opportunity to thoroughly proofread the manuscript to ensure that there are no spelling or grammar issues.

We have thoroughly proofread the manuscript again.

2. Please avoid long steps/notes (more than 4 lines).

We have shortened or split our long protocol steps to avoid going beyond 4 lines where possible.

3. JoVE cannot publish manuscripts containing commercial language. This includes company names of an instrument or reagent. Please remove all commercial language from your manuscript and use generic terms instead. All commercial products should be sufficiently referenced in the Table of Materials and Reagents. Examples of commercial language in your manuscript include Leica, Zeiss, HyD, etc.

We have removed all references to commercial products.

4. Step 5.1.2: Please write this step in the imperative tense.

We have revised the tense of this step.

5. 5.3.14: Please write this step in the imperative tense.

We have revised the tense of this step.

6. Please do not abbreviate journal titles for references.

We have revised our reference list.